Author’s response to reviews

Title: Analysis of Cardiovascular Mortality, Bleeding, Vascular and Cerebrovascular Events in Patients with Atrial Fibrillation vs. Sinus Rhythm Undergoing Transfemoral Transcatheter Aortic Valve Implantation (TAVR)

Authors:

Joerg Herold (Joerg_herold@hotmail.com)
Vasiliki Herold-Vlanti (v1985@windowslive.com)
Mohammad Sherif (mohammad.sherif@med.uni-rostock.de)
Blerim Luani (blerim.Luani@med.ovgu.de)
Klaus Bonaventura (kbonaventura@klinikumevb.de)
Christin Breyer (christinbreyer@gmx.de)
Ruediger Braun-Dullaeus (r.Braun-Dullaeus@med.ovgu.de)

Version: 1 Date: 18 Aug 2017

Author’s response to reviews:

Dear Editor,

The recommended changes were done and the paper was corrected by the American Journal Experts.

Best regards

Joerg Herold